Author's response to reviews

Title: Intervention description is not enough: evidence from an in-depth multiple case-study on the untold role and impact of context in RCTs of 7 Complex Interventions.

Authors:

Mary Wells (e.m.wells@dundee.ac.uk)
Brian Williams (brian.williams@stir.ac.uk)
Shaun Treweek (streweek@mac.com)
Joanne Coyle (joannecoyle2002@hotmail.com)
Julie Taylor (Julie.Taylor@NSPCC.org.uk)

Version: 6 Date: 21 May 2012

Author's response to reviews: see over
Dear Professor Goldsmith,

MS: 5505746621599387 - Intervention description is not enough: evidence from an in-depth multiple case-study on the untold role and impact of context in RCTs of 7 Complex Interventions.

Thank you for your email indicating acceptance of the above paper, pending two minor corrections. Please find attached our revised final manuscript.

We have addressed the two minor corrections as follows: -

1. P 33, p 4, l 1. An extra [.] to read [it...a] has been inserted.
2. P 37, R 30, l 1. “edn” has been deleted.

We have checked that our manuscript complies with all guidelines. We are delighted that you have agreed to accept our paper and look forward to receiving the proofs in due course.

Yours sincerely,

Dr Mary Wells and Prof Brian Williams